Author's response to reviews

Title: The effects of the Two-Week Rule on NHS colorectal cancer diagnostic services: A systematic literature review

Authors:

Kymberley Thorne (k.thorne@swansea.ac.uk)
Hayley A Hutchings (h.a.hutchings@swansea.ac.uk)
Glyn Elwyn (elwyn@cardiff.ac.uk)

Version: 3 Date: 16 March 2006

Author's response to reviews:

Please find enclosed the revised version of the article "The effects of the Two-Week Rule on NHS colorectal cancer diagnostic services: A systematic literature review" (Manuscript ID 1041144741863448) for publication in BMC Health Services Research.

We have altered the inaccuracy highlighted by one reviewer in which we wrongly quoted a figure of 26.1% instead of 9.35% in Table 2. All necessary changes in the text associated with this have been made.

As far as we are aware, no other changes were necessary and we are pleased to offer this submission as our final version.

Thank you for your time.

Mrs Kymberley Thorne